Author’s response to reviews

Title: Protocol for a randomized controlled dismantling study of an Internet-based intervention for depressive symptoms: Exploring the contribution of behavioral activation and positive psychotherapy strategies.

Authors:
Adriana Mira (miraa@unizar.es)
Amanda Díaz-García (amdiaz@uji.es)
Diana Castilla (castilla@uji.es)
Daniel Campos (camosd@uji.es)
Sonia Romero (soniaa.psic@gmail.com)
Bretón-López Juani (breton@uji.es)
García-Palacios Azucena (azucena@uji.es)
Rosa Baños (banos@uv.es)
Cristina Botella (botella@uji.es)

Version: 2 Date: 25 Mar 2019

Author’s response to reviews:

Dear Matthew Hickey,

We appreciate very much your answer. We have submitted the reviewed version of the paper entitled “Protocol for a randomized controlled dismantling study of an Internet-based intervention for depressive symptoms: Exploring the contribution of behavioral activation and positive psychotherapy strategies”, to be considered for publication in BMC Psychiatry, following all your suggestions.

We have revised carefully all the manuscript and duplicated text have been rephrased.
There are some words, as the name of some intervention modules or the name of some intervention program tools, that remain the same, because are proper nouns and it is not possible to change them.

However, if you feel that this version of the manuscript could be improved, please, let us know and we will make the appropriate modifications.

We submitted a clean copy of the manuscript without tracked changes.

Apology for the inconvenience and thank for your consideration.

We are looking forward to receiving your reply.

Sincerely,

Adriana Mira, PhD